Author’s response to reviews

Title: PAC down-regulates Estrogen Receptor alpha and suppresses Epithelial-to-Mesenchymal Transition in breast cancer cells

Authors:

Huda Al-Howail (alhowail@kfshrc.edu.sa)
Hana Hakami (hakami@kfshrc.edu.sa)
Basem Al-Otaibi (alotaibi@kfshrc.edu.sa)
Amer Al-Mazrou (almazrou@kfshrc.edu.sa)
Maha Daghestani (daghestani@kfshrc.edu.sa)
Ibrahim Al-Jammaz (aljammaz@kfshrc.edu.sa)
Huda Al-Khalaf (alkhalaf@kfshrc.edu.sa)
abdelilah aboussehra (aboussehra@kfshrc.edu.sa)

Version: 2 Date: 07 May 2016

Author’s response to reviews:

Response to reviewer’s comments

We are grateful to the reviewers for their relevant and positive comments.

Editorial requirements:

- Please provide email address details for all authors on the title page of the manuscript.

This was added accordingly

- We have recently changed our guidance for authors and we now require a Declarations section. Please include a full Declaration section at the end of the manuscript, which should include all the information as detailed in our Instructions for Authors (please see:}
Declarations were added

- Please ensure that there are authors' contributions for all of the authors. We believe there is one author missing from the section.

This author was added

- Please upload a clean version of the manuscript without tracked changes or highlighting and then remove any highlighted versions from the inventory.

This has been done accordingly